Author's response to reviews

Title: A unique phenotype in a patient with a rare triplication of the 22q11.2 region, and new clinical insights of the 22q11.2 deletion and microduplication syndromes: a report of four cases

Authors:

Sara O Vaz (saravaz87@gmail.com)
Renato Pires (Renato.HM.Pires@azores.gov.pt)
Luís M Pires (lmpires@yahoo.com)
Isabel M Carreira (i_marques@hotmail.com)
Rui Anjos (ranjos@netcabo.pt)
Luisa Mota-Vieira (Luisa.MQ.Vieira@azores.gov.pt)
Paula Maciel (belomaciel@sapo.pt)

Version: 2 Date: 14 November 2014

Author's response to reviews: see over
MS: 2042602960148469 “A unique phenotype in a patient with a rare triplication of the 22q11.2 region, and new clinical insights of the 22q11.2 deletion and microduplication syndromes: a report of four cases”

CARE checklist

<table>
<thead>
<tr>
<th>Topic</th>
<th>Item</th>
<th>Authors corrections</th>
</tr>
</thead>
<tbody>
<tr>
<td>Title</td>
<td>1.</td>
<td>The Checklist points that the words “case report” should be in the title along with the area of focus. We revised the manuscript accordingly by adding both words (please, see Title Page, page 1, line 3).</td>
</tr>
</tbody>
</table>

Ponta Delgada, November 14th, 2014

Luisa Mota-Vieira, PhD

Head of Department
UGPM – Molecular Genetics and Pathology Unit
Tel. (+351) 296 203 531
Fax. (+351) 296 203 090
e-mail: Luisa.MQ.Vieira@azores.gov.pt